Centre for Global Finance

Working Paper Series

No.16 / 2019

Segmentation of the interbank money market in Zambia

By Jonathan M Chipili, Francis Z Mbao, Alick B Lungu, Shula M Sikaona, Anthony Bwalya, Cosam S Chanda



The Centre for Global Finance (CGF) Working Paper Series features recent studies by resident members of CGF as well as visiting researchers, altogether demonstrating the depth and breadth of research being undertaken at CGF. The papers are published to facilitate preliminary dissemination of ongoing research, enhance quality of work and contribute to the advancement of knowledge. We acknowledge, without implication, financial support from the DEGRP Research Grant (ES/N013344/2), funded by DFID and ESRC, on "Delivering Inclusive Financial Development and Growth", the ESRC-NSFC (ES/P005241/1) Research Grant on "Developing financial systems to support sustainable growth in China – The role of innovation, diversity and financial regulation", and the AXA Research Fund.

List of previous Working Papers of CGF:

No.1/2018	Capital, risk and profitability of WAEMU banks: Does cross-border banking matter? By Désiré Kanga, Victor Murinde, and Issouf Soumaré
No.2/2018	<i>Capital flows and productivity in Africa: The angel is in the details</i> . By François A. B. Bationo, Stephany Griffith-Jones, Victor Murinde, Issouf Soumaré and Judith Tyson
No.3/2018	<i>The persistence of bank fragility in Africa: GMM dynamic panel data evidence.</i> By Abbi M. Kedir, Syed Faizan Iftikhar, Victor Murinde and Bernadette Kamgnia
No.4/2018	Reflections on central banking. By Victor Murinde and Patrick Njoroge
No.5/2018	Let beholders behold: Can banks see beyond oil booms and mitigate the Dutch disease? By Morakinyo O. Adetutu, John E. Ebireri, Victor Murinde and Kayode A. Odusanya
No.6/2018	<i>National culture, CEO power and risk-taking by global banks</i> . By Eilnaz Kashefi Pour and Victor Murinde
No.7/2018	Corporate investment, financing and payout decisions under financial constraints and uncertainty: Evidence from UK firms. By Qingwei Meng, Victor Murinde and Ping Wang
No.8/2018	Bank opacity and risk-taking: Evidence from analysts' forecasts By Samuel Fosu, Collins G. Ntim, William Coffie, and Victor Murinde
No.9/2018	Does microcredit increase hope, aspirations and well-being? Evidence from Sierra Leone. By Adriana Garcia, Robert Lensink, and Maarten Voors
No.10/2018	Lessons from emerging economies for African low income countries on managing capital flows. By Stephany Griffith-Jones and José Antonio Ocampo
No.11/2018	Financial inclusion and economic growth: What do we know? By Joshua Y. Abor, Haruna Issahaku, Mohammed Amidu, and Victor Murinde

No.12/2018	<i>Climate vulnerability and the cost of debt.</i> By Gerhard Kling, Yuen C Lo, Victor Murinde, and Ulrich Volz
No.13/2018	Pan-African banks on the rise: Does cross-border banking increase firms' Access to finance in WAEMU? By Désiré Kanga, Victor Murinde, Lemma Senbet, and Issouf Soumaré
No.14/2018	The peer monitoring role of the interbank market and implications for bank regulation: Evidence from Kenya. By Victor Murinde, Ye Bai, Christopher J. Green, Isaya Maana, Samuel Tiriongo, and Kethi Ngoka-Kisinguh
No.1/2019	Central bank independence: What are the key issues? By Désiré Kanga and Victor Murinde
No.2/2019	Banking services and inclusive development in sub-Saharan Africa. By Haruna Issahaku, Mohammed Amidu and Aisha Mohammed Sissy
No.3/2019	A survey of literature on financial literacy and financial behaviour: Is there a gender gap? By Maryam Sholevar and Laurence Harris
No.4/2019	Capital adjustment over the cycle: Evidence from microfinance institutions. By Issouf Soumaré, Hubert Tchakoute Tchuigoua, and Hélyoth T.S. Hessou
No.5/2019	Provisioning and business cycle: Evidence from microfinance institutions. By Hélyoth T.S. Hessou, Robert Lensink, Issouf Soumaré, and Hubert Tchakoute Tchuigoua
No.6/2019	Lending and business cycle: evidence from microfinance institutions. By Hubert Tchakoute Tchuigoua, Issouf Soumaré, and Hélyoth T.S. Hessou
No.7/2019	Term structure of CDS spreads & risk-based capital of the protection seller: an extension of the dynamic Nelson-Siegel model with regime switching. By Standley R. Baron and Issouf Soumaré
No.8/2019	<i>Confidence, financial inclusion and sustainable economic development.</i> By Ayse Demir, Reinhard Bachmann, Victor Murinde, Laurence Harris, Christine Oughton and Meng Xie
No.9/2019	The network structure of the Malawi interbank market: implications for liquidity distribution and contagion around the banking system. By Esmie Koriheya Kanyumbu
No.10/2019	Aid and Exchange Rates in sub-Saharan Africa: No More Dutch Disease? By Oliver Morrissey, Lionel Roger and Lars Spreng
No.11/2019	Does credit deepening increase financial fragility? By Peng Yiqing, Niels Hermes, and Robert Lensink
No.12/2019	Does microcredit increase aspirational hope? Evidence from a group lending scheme in Sierra Leone. By Adriana Garcia, Robert Lensink, and Maarten Voors

No.13/2019	Do better formal institutions promote financial inclusion? By Peng Yiqing, Niels Hermes, and Robert Lensink
No.14/2019	Do interbank interest rates reflect the financial soundness of borrowing banks? By Thomas Bwire, Martin Brownbridge, Doreen K. Rubatsimbira and Grace A. Tinyinondi
No.15/2019	Institutional environment and the microstructure of the interbank market. By

No.15/2019 Institutional environment and the microstructure of the interbank market. By Thomas Bwire, Martin Brownbridge, Doreen K. Rubatsimbira, and Grace A. Tinyinondi

Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorisation of the authors of this paper. The views expressed in the paper are those of the authors and do not necessarily reflect those of the CGF.

All CGF Working Papers can be downloaded from CGF Website.

Centre for Global Finance SOAS University of London 10 Thornhaugh Street, Russell Square London WC1H 0XG

Email: cgf@soas.ac.uk Website: <u>https://www.soas.ac.uk/centreforglobalfinance/publications/</u>

SEGMENTATION OF THE INTERBANK MONEY MARKET IN ZAMBIA

Jonathan M Chipili^{*}, Francis Z Mbao^{*}, Alick B Lungu^{*}, Shula M Sikaona^{*}, Anthony Bwalya^{*}, and Cosam S Chanda^{*}

Abstract

This study examined segmentation in the interbank money market. Commercial banks were classified according to asset size and ownership. Network framework analysis was used to establish lending and borrowing preferences as well as the utilization of credit lines in the market. Segmentation by pricing behavior was also undertaken by assessing the deviation of the interbank rate from the policy rate for trades within and across bank categories. Further, long-range dependence in interbank pricing was conducted by estimating the Hurst exponent. Daily data spanning from January 2012 to March 2019 were used. Preliminary results have revealed the existence of segmentation in the interbank market: the market structure is incomplete (not all banks have credit lines), but not disjointed as both within and cross bank categories trades occur to some extent. Specifically, large banks fully trade with each other by utilizing all the available credit lines. Nonetheless, they charge a relatively higher premium than small and medium banks. The interaction among small banks and peers is relatively high and attracts a relatively lower premium than is the case with lending to medium and large banks. There is limited interaction among the medium-medium pair and tend to charge each other a relatively higher premium. *Ownership tends to matter in interbank trading. Bank categories with similar or related ownership* tend to have more interactions as opposed to bank categories with diverse ownership. All the trading pairs of banks considered transacted above the policy rate and the deviation was even higher when monetary policy was tightened. Finally, there is evidence of uniform pricing behavior in the interbank money market and predictability in interbank lending rates, suggesting absence of price segmentation. However, the pricing behavior is susceptible to monetary policy stance change which makes lending rates unpredictable for some trading relationship pairs.

1 Introduction

The interbank market was in its nascent stage prior to 1990 and dominated by banks¹ established after Zambia gained independence in 1964. The financial sector was dominated by foreign commercial banks mainly serving the needs of expatriate and foreign businesses (Brownbridge, 1996). The market was characterized by excess liquidity and there is very little evidence of interbank money market trading in the early post-colonial years (Bank of Zambia Annual Report, 1968 and 1977). However, as the market started to develop from the 1970s, economic reforms

^{*}Bank of Zambia, P.O. Box 30080, Lusaka, 10101. Zambia. Corresponding author: Jonathan M Chipili. E-mail: jchipili@boz.zm Tel: +260 211 399 300. The views expressed in this paper do not in any way represent the official position of the Bank of Zambia. Chipili remains responsible for all the errors and omissions. We acknowledge, without implication, financial support from the DEGRP Research Grant (ES/N013344/2), funded by DFID and ESRC, on "Delivering Inclusive Financial Development and Growth".

¹ Only four commercial banks established in the 1960s exited.

adopted in the early 1990s, and with the increase in the number of commercial banks, the importance of interbank money market became more pronounced in terms of the traded volumes.

By 1994, there were signs of the financial system transitioning towards market-based structures as exchange controls were abolished, interest rates decontrolled, control on goods prices lifted, and the Banking and Financial Services Act enacted in 1994. In addition, secondary market trading in government securities, not only between commercial banks but also between non-bank financial institutions, emerged. During this period, a number of commercial banks started to rediscount government securities while repurchase agreements involving Treasury bills between commercial banks and their clients also increased (Bank of Zambia Annual Report, 1994). With these developments, interbank money market transactions rose and overnight loans reached K40 billion within weeks from virtually no trades at all (Bank of Zambia Annual Report, 1994).

The interbank money market in Zambia is dominated by commercial banks, although there are other institutions such as pension funds, building societies, insurance companies, a development bank, a national savings and credit bank that borrow or lend funds in the overall financial sector. Treasury bills are widely used as collateral in interbank transactions. However, to date, there are no clearly defined and specific laws or regulations to guide the interbank money market, let alone a code of conduct for money market players. The code of conduct guides market participant's responsibilities, integrity, trust, honesty and faith in dealing with interbank transactions.

The interbank market is indirectly regulated by some provisions of the Bank of Zambia Act and the Banking and Financial Services Act by virtue of this market being dominated by commercial banks. In addition, the interbank money market benefits, to a large extent, from spillover regulations governing the foreign exchange and government securities markets. Unlike the foreign exchange market where participants provide both bid and offer prices using a trading platform, participants in the interbank money market trade with each other over-the-counter (OTC) where banks negotiate among themselves.

Typically, commercial banks tend to exhibit different behavior in the interbank market based on their size. There are several approaches to classifying a bank as large, medium or small. This could be based on different quantiles from bank asset size distribution (Kim, 2017). In Zambia, the majority of small banks dominate on the borrowing side while big banks tend to be mostly on the lending side (Muhanga, Mutoti and Zgambo, 2009). In addition, large banks have a huge deposit base which they exploit to provide funds to small banks. However, not all the large banks are active in the interbank market mostly due to their internal policies. Further, overnight transactions dominate the trades in the interbank money market on both secured and unsecured basis depending on the relationship with the parties involved (Muhanga, Mutoti and Zgambo, 2009). However, while the overnight interbank market is the most active, structural constraints or rigidities exist that contribute to the segmentation of the market.

Segmentation prevents the interbank money market from effectively performing its typical function of optimally distributing liquidity, facilitating the transmission of monetary policy, as well as stabilising and contributing to the efficiency of the financial system (Oduor *et al.*, 2014; Mayordomo *et al.*, 2015; and Osoro and Muriithi, 2017). Segmentation can be measured in terms of prices and volumes. Typically, market segmentation should lead to high dispersion in market

prices or interest rates quoted by participants which makes it hard for them to infer a fair price (Mayordomo *et al.*, 2015).

While the volume of interbank trading has increased over the years (Figure 1 in the appendix), structural rigidities are cited as contributing to the persistent deviation of the overnight interbank rate (operating target since April 2012)² from the monetary policy rate in Zambia (Figure 2 in the appendix). As a consequence, banks continue to rely on central bank standing liquidity facilities for adjustment funds (Figure 3 in the appendix) despite the interbank market being characterized by excess liquidity (Figure 4 in the appendix). In addition, it is observed in figure 4 that there are periods when the overnight interbank rate counterintuitively responds to changes in liquidity conditions as opposed to the traditional inverse relationship (Lungu, Bwalya and Chanda, 2018). Gereben (1999) characterizes this as an anomaly of the interbank money market. Persistent volatility in the interbank rate, irrespective of the underlying factors, tends to affect other interest rates with longer maturities which in turn creates uncertainty, constrain consumer and business investment spending and ultimately economic growth. Further, in the presence of distortions, the market is not able to precisely distinguish the intended effects of the monetary policy stance through changes in the target policy rate and changes induced by liquidity shocks. Commercial banks respond to liquidity shocks by borrowing in the interbank market or liquidating assets. However, borrowing tends to be limited by credit lines and when it is possible, stringent borrowing constraints exist. Large banks are reluctant to create sufficient credit lines with small banks partly due to mistrust (Arukaevu, 1998). Commercial banks with higher levels of credit risk tend to suffer the most during periods of severe market stress when banks are not willing to lend to each other.

Interest rates vary across different lender-borrower categories or pairs undertaken under mutual agreement. Underlying this difference is the liquidity cost³, default or credit risk, and strength of relationship among lender-borrower pair. The marginal cost of acquiring funds during periods of liquidity shortage increases with the size of the shortage and the marginal cost of offloading excess funds in an attempt to earn a return increases with the accumulated amount of excess liquidity (Kim, 2017). The liquidity cost for large banks is zero as parties are not willing to accept a lower return on excess liquidity as they are not willing to accept a higher borrowing cost (Kim, 2017). This results in little variation in interest rates such that trading between large banks tends to be very close to the central bank target (policy) rate. Kim (2017) found that small banks tend to lend funds to large banks below the central bank target rate; small banks tend to borrow at interest rates above the central bank target rate from large banks; and favourable rates are applied when a small bank trades with a large bank for larger loans: receive a higher rate for big loans provided to large banks and get charged a lower rate when borrowing from a large bank for a large loan.

Thus, various aspects of the interbank market are analysed, including the drivers of the cost of funds, and the effect of the state of financial system on the outcome of the interbank market (Kim, 2017). This study analyses the extent, and not the underlying factors, of segmentation of the interbank money market in Zambia. It extends Muhanga, Mutoti and Zgambo (2009) and examines interbank market segmentation by covering a longer sample period (2012m1-2019m3), and

² The Bank of Zambia shifted from a monetary targeting to an interesting rate targeting framework in April 2012. The operating target changed to the overnight interbank rate from base money. With this framework, the Bank of Zambia seeks to steer the interbank rate as close as possible to the target rate (policy rate) within the defined band.

³ How banks handle the cost of excess or deficit liquidity.

broaden the classification of banks by focusing on bank total assets, ownership, volumes, and prices using network framework analysis, lending and borrowing (trading) preference indices, as well as price differential analysis. Muhanga, Mutoti and Zgambo (2009) focused on a relatively smaller sample (2006-2009), solely on ownership structure of banks (domestic or foreign) of small and big banks, and price dynamics analysis. Understanding the role of medium sized banks in the interbank market is equally important than just the small and large (big) banks as they may help to establish the category of banks that may pose (in) efficiencies in the market. Inefficiencies in the interbank market tend to impede the smooth operation of monetary policy.

Preliminary results have revealed the existence of segmentation in the interbank market: the market structure is incomplete (not all banks have credit lines), but not disjointed as both within and cross bank categories trades occur to some extent. Specifically, large banks fully trade with each other by utilizing all the available credit lines. Nonetheless, they charge a relatively higher premium than small and medium banks. The interaction among small banks and peers is relatively high and attracts a relatively lower premium than is the case with lending to medium and large banks. There is limited interaction among the medium-medium pair and tend to charge each other a relatively higher premium. Ownership tends to matter in interbank trading. Bank categories with similar or related ownership tend to have more interactions as opposed to bank categories with diverse ownership. All the trading pairs of banks considered transacted above the policy rate and the deviation was even higher when monetary policy was tightened. Finally, there is evidence of uniform pricing behavior in the interbank money market and predictability in interbank lending rates, suggesting absence of price segmentation. However, the pricing behavior is susceptible to monetary policy stance change which makes lending rates unpredictable for some trading relationship pairs.

The rest of the paper is organized as follows: Section 2 presents the literature on interbank money market segmentation while section 3 outlines the methodology. Section 4 discusses data sources. Section 5 presents preliminary findings and section 6 concludes.

2 Literature Review

The interbank market plays a critical role of allocating liquidity from banks with surplus funds to banks facing liquidity deficits. The ability of banks to trade in the interbank market is principally affected by liquidity shocks, their operating costs and reputation as well as the operations of the central bank (Green *et al.*, 2016). The interbank money market is mainly utilized as a source of funds for short-term liquidity obligations and not to expand liquidity. Anticipation of changes in policy and/or operating costs influence individual bank's trading decisions as well as decisions to supply or withhold liquidity in the interbank market.

Shocks caused by liquidity stress may generate distortions and inefficiencies in the functioning of the interbank market including the cost of funds which may be transacted in a characteristically segmented market. The interbank money market may be segmented in terms of products offered, pricing, concentration of liquidity, and the operational structure in which banks trade with each other (Osoro and Muriithi, 2017).

Sichei, Kiplang and Shimba (2012) demonstrate how segmentation can limit the interbank market's ability to facilitate banks' liquidity management strategy in the Kenyan interbank market. The study employed the network framework analysis and case studies of countries with developed interbank markets, namely the UK, EU and the US. Daily data for the Kenyan interbank market covering the period June 2003-September 2012 were used. The interbank market was found to be incomplete and highly segmented by size: small, medium and large banks. Large banks tended to discriminate against small banks in terms of credit extension and the interest rate charged which was usually higher than that charged on their peers. The study concluded that segmentation was the reason the interbank market in Kenya had limited ability to facilitate banks' liquidity management strategy. Further, the analysis of the case studies revealed that the efficiency of the interbank market in Kenya could be enhanced through a number of developments such as adding lending products with maturities of more than one day to extend the term structure, increasing the number of currencies traded, developing a benchmark interbank interest rate, and increasing linkages with other money market segments and monetary policy.

Muhanga, Mutoti and Zgambo (2009) investigated segmentation in the interbank money market in Zambia during the period 2006-2009. The study focused on the distribution of funds in the market based on pricing mechanisms using period averages and comparative borrowing costs. The findings revealed segmentation of the interbank market both in terms of the distribution of funds and pricing. On average, 42.7% of the settlement balances maintained at the central bank was held by major foreign-owned banks while an average proportion of 42.4% was held by two major locally-owned banks. With regard to pricing, the results showed that locally-owned smaller banks tended to borrow funds at interest rates usually higher than the average market rate while foreignowned large banks tended to borrow below market averages. The study concluded that the concentration of funds among a few banks would be the basis for market segmentation as the dominant banks in the interbank market may use their position to influence the pricing of interbank funds.

Relationships and network dynamics also appear to increasingly influence interactions in the interbank money market (Green, et al, 2016). The position and direction of the credit relationship between banks seems to have an effect on the price a bank pays for liquidity in the interbank market (Craig, Fecht and Tümer-Alkan, 2015; and Temizsoy, Iori and Montes-Rojas, 2015). Studies by Chiu, Eisenschmidt and Monnet (2019) Colliard *et al.*, (2016) and Bräuning and Fecht (2017) all concluded that banks rely on repeated interactions with the same counterparties to access liquidity. Banks choose to build relationships in order to insure against liquidity shocks and to economize on the cost to trade in the interbank market. They further claimed that relationships explain some anomalies in the level of interest rates and how monetary policy affects the network structure of the interbank market and its functioning.

Large banks prefer to lend to each other at rates close to the central bank target rate and limit their transactions with small banks as they view them as very risky due to the size of their assets (Allen and Gale, 1990; and Allen and Saunders, 1992). Thus, small banks are left with the option of borrowing funds from their peers which may not be adequate (Ho and Saunders, 1985). Small banks borrowing from large banks tend to pay higher rates than the central bank target rate. However, they get lower rates than the central bank target rate when they lend to large banks. It follows therefore that large banks usually tend to be net borrowers while smaller banks tend to be

net lenders, the principle at the core of the 'small-bank' – 'big-bank' dichotomy concept initially advanced by Ho and Saunders (1985).

Kim (2017) draws on the 'small-bank' – 'big-bank' dichotomy concept and contends that empirically, odds point to outcomes being against small banks partly attributed to information asymmetry between a small borrowing financial institution and a prospective counterparty. In addition, some banks tend to shy off from banks with less profitable and lower market reputation.

The foregoing depicts a high degree of continuous interaction and interdependence with connections among banks stemming from both the asset and the liability sides of their balance sheets which has evolved in a web of structured relationships increasingly classified as networks – the core of the network theory. Formerly, Chen, Zhang and Li (2016) defined a network as a collection of nodes and edges that can be a useful method to depict the interrelationships among organizations and explain the idiographic effect that one organization has on another. Each bank is taken as a node and the credit lending relationship between banks as the edges connecting these nodes in the network. Thus, the nodes and edges compose the interbank market network. In these networks, nodes in each category for example "small and big" or "local and foreign" classes generally prefer to link to the nodes.

Colliard, Foucault and Hoffmann (2016) studied the core-periphery model of trading in the overnight interbank market during crisis period in the euro area. The study focused on the market power of periphery banks connected to the core, dispersion of rates in the interbank market and the inefficient recourse to the central bank standing facilities. They concluded that segmentation between core and periphery banks poses significant challenges to central banks. They found that there is dispersion in interest rates between core and periphery banks that no longer reflect borrowing conditions of the interbank market.

Craig, Fecht and Tümer-Alkan (2015) assessed the interconnectedness of banks and the price they pay for liquidity in Germany. The concentration of credit relationships and the position of a bank in the network topology of the financial system influence the bank's ability to meet liquidity demand in Germany. Controlling for bank characteristics and daily fulfilment of reserves requirements, the study findings revealed that banks with a more diversified borrowing structure in the interbank market bid significantly less aggressively and pay a lower price for liquidity in the ECB's main refinancing options. Further, Gabrieli (2012) investigated the overnight lending market network in European countries. The study established that the network is very scattering, with characteristics of a small-world network, and the distribution of its nodes subject to power-law distribution. This implies that the majority of banks only build relationships with fewer banks and there are only a few banks (usually large banks) that have a larger network connection. Similar to the European bank overnight market network, the federal funds market network also has the characteristics of small-world network, but the distribution of network node degree is a heavy-tailed rather than a power-law distribution (Bech and Atalay, 2010).

3 Empirical Methodology

To investigate the extent of interbank money market segmentation in Zambia, the study employs network framework analysis, lending and borrowing (trading) preference indices, as well as price differential analysis.

As alluded to in section 1, the assessment of interbank segmentation takes into account the size of the bank. In this study, banks are classified according to asset size as follows:

a) Determine the ratio of an individual bank total asset to industry total asset using balance sheet data:

$$\beta_{it} = \left(\frac{\theta_{it}}{\Omega_t}\right) 100\%, i = 1, \dots, \dots, \dots, 18 - - - - - - (1)$$
$$t = 0, 1, 2, \dots, \dots, T$$

where

 $\beta = Bank size$ $\theta = Individual bank total assets$ $\Omega = Industry total assets$

b) Compute the average bank size for each individual bank for the period January 2017-February 2019 when closures and mergers of banks took place:

where

$$\beta_i = Mean \ bank \ size \ for \ individual \ banks$$

 $(T - t) = Number \ of \ observations$

c) Decision rule for a bank being either large, medium or small is as follows:

$$\bar{\beta}_i = \begin{cases} \ge 0.10, \text{large bank} \\ 0.03 \ but < 0.10, medium - - - - - - - (3) \\ < 0.03, \text{small bank} \end{cases}$$

If the average market share is at least 10 percent (0.10), such a bank is considered large. Otherwise it is small if the market share is less than 4 percent (0.03), and medium if the market share is between at least 4 and less than 10 percent. The rule is arbitrarily determined, but ensures that large banks have a combined market share of at least 50 percent share of the industry; small banks have a combined market share; and for medium size banks, their market share should be in the middle of the pack.

Applying the procedure to the data involving 14 commercial banks, five banks are classified as small with the average market share ranging from 1.15 to 1.63 percent (Table 1). The combined market share is only 6.98 percent. There are five medium sized banks with the market share ranging from 3.16 to 9.68 percent, and combined market share of 37.56 percent. Four large banks are

identified with the market share ranging from 12.17 to 16.58 percent, and the total market share of 55.46 percent.

Banks are also classified based on ownership within each category. Ownership is one of the factors considered when setting credit lines among banks. It is therefore imperative to understand how ownership affects interbank trades within and across different bank categories. When setting credit lines, at the group or board level, most banks check for bank ownership as a default risk measure (Lungu, Chanda, & Bwalya, 2018).

Bank	Markets Share	Classifica	ation	Ownership
	(%)			
Bank S1	1.4	Small		Local
Bank S2	1.6	Small		Local
Bank S3	1.2	Small		Local
Bank S4	1.5	Small		Local
Bank S5	1.3	Small		Foreign
Bank M1	7.6	Medium		Foreign
Bank M2	9.7	Medium		Foreign
Bank M3	3.6	Medium		Foreign
Bank M4	3.2	Medium		Foreign
Bank M5	8.4	Medium		Local
Bank B1	13.8	Large		Foreign
Bank B2	12.2	Large		Foreign
Bank B3	16.6	Large		Foreign
Bank B4	12.9	Large		Local

Table1 Bank Classification by Asset Size

Source: Author's computation using Bank of Zambia database

Network Framework Analysis

The network framework proposed by Allen and Gale (2000) is based on interbank credit lines in which exposure matrices are used. According to this framework, three types of interbank structures exist: complete, incomplete as well as incomplete and discounted.

A complete structure is where each bank is symmetrically connected to other banks in the market irrespective of their size (Table 2). This means that each bank transacts (borrows and lends) with all the banks in the market (Sichei, Kiplang and Shimba, 2012). When an interbank market has a complete structure, the effect caused by unexpected shock in one bank can be absorbed by a large number of banks thereby reducing the intensity of the shock (Allen and Gale, 2000).

			Borrowing										
			Small Bank	(S	N	Medium Banks			Large Banks				
		Bank S1	Bank S2	Bank S3	Bank M1	Bank M2	Bank M3	Bank L1	Bank L2	Bank L3			
		Bank S1		٧	٧	٧	٧	٧	٧	٧	٧		
	Small Banks	Bank S2	٧		v	٧	٧	٧	v	٧	٧		
		Bank S3	٧	٧		٧	V	V	v	V	٧		
		Bank M1	v	٧	٧		V	V	٧	V	V		
Lending	Medium Banks	Bank M2	٧	٧	٧	٧		٧	٧	٧	٧		
		Bank M3	V	٧	٧	٧	٧		٧	٧	٧		
		Bank L1	٧	٧	٧	٧	٧	٧		٧	V		
	Large Banks	Bank L2	٧	٧	٧	٧	٧	٧	V		V		
		Bank L3	٧	٧	٧	٧	٧	٧	٧	٧			

Source: Author's computations

 $\sqrt{}$ means existence of an active interbank credit line while 0 implies no trade between the corresponding banks

Note: S- small banks, M-medium sized banks and L-large banks

In incomplete interbank markets, banks are only connected to their neighbours i.e. those in the same or neighbouring segment. For instance, small banks trade amongst themselves and medium banks, but do not trade with large banks (Table 3). When the interbank market is incomplete, the initial shock in one bank is transmitted to its neighbours, but in a large magnitude with ripple effects.

	-		_				Borrowing	5				
		Small Banks					Medium Banks			Large Banks		
			Bank S1	Bank S2	Bank S3	Bank M1	Bank M2	Bank M3	Bank L1	Bank L2	Bank L3	
		Bank S1		v	٧	v	٧	٧	0	0	0	
	Small Banks	Bank S2	٧		٧	٧	٧	٧	0	0	0	
		Bank S3	٧	٧		٧	٧	٧	0	0	0	
		Bank M1	٧	V	٧		V	٧	V	V	٧	
Lending	Medium Banks	Bank M2	v	v	٧	٧		٧	V	V	V	
		Bank M3	٧	V	٧	V	٧		V	V	٧	
		Bank L1	0	0	C	V	٧	٧		V	V	
	Large Banks	Bank L2	0	0	C	v	v	٧	V		V	
		Bank L3	0	0	C	V	٧	٧	V	V		

Table 3 Incomplete Structure

Source: Author's computations

 $\sqrt{}$ means existence of an active interbank credit line while 0 implies no trade between the corresponding banks

Note: S- small banks, M-medium sized banks and L-large banks

In an incomplete and disconnected interbank market structure, trades only occur within bank categories i.e. there are no cross category trades. (Table 4).

			_		*		Borrowing					
			9	Small Bank	S	М	edium Ban	ks	Large Banks			
		Bank S1	Bank S2	Bank S3	Bank M1	Bank M2	Bank M3	Bank L1	Bank L2	Bank L3		
		Bank S1		v	V	0	0	0	0	0	0	
	Small Banks	Bank S2	٧		v	0	0	0	0	0	0	
		Bank S3	٧	V		0	0	0	0	0	0	
		Bank M1	0	0	0		V	٧	0	0	0	
Lending	Medium Banks	Bank M2	0	0	0	V		V	0	0	0	
		Bank M3	0	0	0	V	٧		0	0	0	
		Bank L1	0	0	0	0	0	0		V	V	
	Large Banks	Bank L2	0	0	0	0	0	0	V		V	
		Bank L3	0	0	0	0	0	0	V	V		

Table 4 Incomplete and Discounted Structure

Source: Author computations

 \sqrt{means} existence of an active interbank credit line while 0 implies no trade between the corresponding banks

Note: S- small banks, M-medium sized banks and L-large banks

Further centrality measures, namely betweenness centrality, closeness centrality, and the cluster coefficient are employed as part of network analysis. Centrality measures in network analysis give a deeper understanding of how interrelationships among banks work.

Under the centrality concept, a network is defined as a collection of points or "nodes" connected together by lines or "edges." The interpretation of nodes and edges depends on the context. In this study, nodes are commercial banks and edges are interbank loans extended to counterparties. A network is a representation of how elements are related in a system, which can be in matrix (exposure matrices highlighted above) or graphical form (Table 5).



 Table 5 Representation of a Network

Source:" Comesa Monetary Institute, 2019

Edges are lines that show relationships between vertices; financial networks assume different edges, depending on what they depict; and Vertices/Nodes are number of items, pieces, banks, countries in a relationship graph.

As earlier highlighted, the interbank money market may be characterized by banks that may not actively transact with the rest. In the context of network analysis, the clustering coefficient can be used to measure segmentation in the interbank market by establishing how connected vertices are to one another. More specifically, it is the number of edges connecting a vertex's neighbours divided by the total number of possible edges between the vertex's neighbours (COMESA Monetary Institute, 2019). If all neighbours of a node are not connected to each other, the coefficient will take the value of 0. Conversely, if all the neighbours are connected, the cluster coefficient will be 1. Thus, the clustering coefficient ranges from 0 to 1 ($0 \le x \le 1$).

Closeness centrality shows the importance of a bank in the network by how close it is it to the counterparties and betweenness measures the importance of the bank by determining its role as a mid-agent between banks without credit lines in the network⁴. Closeness centrality in this study indicates how close a commercial bank is to all the banks in the network in terms of interbank trades. For betweenness centrality, the measure captures a commercial bank's role in acting as a mid-agent in liquidity flow between two banks with no credit lines in the network. Commercial banks may act as a mid-agent to facilitate the flow of liquidity between two of its counterparties who do not directly have credit lines despite both having credit lines with it.

⁴ <u>https://www.sciencedirect.com/topics/computer-science/betweenness-centrality</u>

Lending and Borrowing (Trade) Preference Indices

To assess the intensity of both lending and borrowing between bank categories, the lender preference index (LPI) and borrower preference index (BPI) are computed in line with Cocco, Gomes and Martins (2009). In this study, banks are categorized into three groups: large, medium, and small based on their market share in relation to total assets. In this case, for every lender category and borrower category, the LPI is computed as follows:

 $LPI_{category \ X \ banks \ to \ category \ Y \ banks} = \frac{Total \ lending \ of \ funds \ by \ category \ X \ banks \ to \ category \ Y \ banks}{Total \ lending \ of \ funds \ by \ category \ X \ banks \ in \ the \ market}$

This ratio is more likely to be high if category X banks rely on fellow category X banks more than they do on category Y banks to lend funds in the market.

The Borrower Preference Index (BPI) is computed in a similar way as follows:

ם –	Total borrowed funds by category X banks from category Y banks
DF1category X banks to category Y banks ⁻	Total borrowed funds by category X banks in the market

Price Differential Analysis

Two measures of pricing behaviour (within and across bank segments) are employed: deviation of interbank lending rate from the central bank target rate and long-range dependence.

It is generally established that interest rates vary across different lender-borrower categories or pairs undertaken under mutual agreement (Kim, 2017). Underlying this difference is mainly the liquidity cost (how banks handle the cost of excess or deficit liquidity), default or credit risk, and the strength of the relationship among lender-borrower pair. Trading between large banks tends to be very close to the central bank target (policy) rate; small banks tend to lend funds to large banks at interest rates below the central bank target rate; small banks tend to borrow at rates above the central bank target rate from large banks; and favourable rates are applied when a small bank trades with a large bank for larger loans⁵(Kim, 2017). In this way, small banks will tend to transact at rates different from the central bank target. Thus, the mean is used to assess the pricing behavior in the money market over time within and across different categories of banks measured as the deviation of the actual lending rate from the central bank target (policy rate).

Long-range dependence techniques are used to establish whether pricing behaviour differs and/or is predictable within and across the segments of the interbank market. If the pricing behaviour is the same, this points to herd behavior, suggesting lack of pricing segmentation. In this study, the Hurst parameter (*H*) is used to measure long-range dependence⁶ (persistence) in lending rates within and across different categories of banks. The parameter values of *H* are defined as $H \in [0,1]$, where 0 < H < 0.5 means the data process is anti-persistent, akin to being mean reverting within some bound - implying a short memory. For 0.5 < H < 1, the process is persistent with a

⁵ Receive a higher rate for big loans provided to large banks and get charged a lower rate when borrowing from a large bank for a large loan.

⁶ In this context, long-range dependence is taken as a characterisation of the underlying process of a given time series.

long memory. For H = 0.5, the process is independent, akin to being a random walk. If the value for the estimated H parameter is within the same range for all the categories of banks, then the pricing behavior should be the same signifying absence of price segmentation in the interbank market.

In this study, the discrete wavelets transform (DWT) is adopted to estimate H as the data on interbank pricing is measured on a discrete basis and is irregular⁷. With the DWT, a given signal is decomposed into filtered series at different time scales.

Long-range dependence in interest rates has been explored by Tabak and Cajueiro (2005), Cajueiro and Tabak (2007), and Cajueiro and Tabak (2009). Although the focus was on establishing long-range dependence in the context of predictability, persistence in behavior, and implications on tests for expectation hypothesis, the idea can also be extended to establishing pricing segmentation. The Hurst parameter is applied in assessing pricing behaviour across market segments and thereby draw insights on different pricing behaviour intra and across bank categories.

Further, data for each trading pair was subjected to structural break tests using an empirical fluctuation process implemented in **R** under the package **strucchange** credited to Zeileis et al. (2002). The rationale behind structural breaks reflect specific events relating to monetary policy decisions as outlined in Table 6 (a-c). As outlined earlier, banks tend to use the central bank policy rate as reference in the pricing of funds as they trade with each other. Thus, structural break tests are intended to determine whether changes in the monetary policy stance influences the pricing behavior of banks in the interbank market. According to Table 6, five monetary policy events were identified and subjected to an empirical test to determine whether they had influence on interbank rate setting. The events mostly relate to changes in the monetary policy rate and the to the overnight lending facility (OLF)⁸ rate.

⁷ Rea et al. (2013) have evaluated the efficacy of various methods used in estimating the Hurst coefficient. They include Fourier spectral techniques, Wavelet, aggregated variance methods, detrended fluctuation analysis (DFA), generalised Hurst exponents (GHE), detrended moving average (DMA), Local Whittle estimator, and the rescaled range analysis (R\S). The Wavelet transform methods have become popular in the estimation of the Hurst parameter (Chamoli, et al., 2007; Kantelhardt, 2009; Kirichenko et al. 2011). The Wavelet was unbiased for all the parameter values for longer lengths, but was minimally (very small) biased for shorter lengths. In this regard, the Wavelet emerged as the relatively more effective method. Kirichenko et al. (2011) also assessed the most commonly used methods in estimating the Hurst coefficient. The detrended fluctuation analysis (DFA) and the Wavelets methods were found to have minimal bias for stationary series, but the DFA was accurate for the non-stationary series while the Wavelets was good also but for the data with a slight trend.

⁸ The overnight lending facility is a discount window from which commercial banks borrow from the central bank for liquidity management. The interest rate (OLF rate) is set off the monetary policy rate on which a margin is added s determined by the central bank.

	Event	Period Covered			Observations	Structural Break Dates	Possible Factors
	1	April 3, 2012	to	March 18, 2014	244	NA	
Small to Small sized Inter-Bank Market	2	March 21, 2014	to	November 4, 2015	338	March 21, 2014	OLF tightened by 600 bps and acess to OLF window restricted to once a week
	3	November 5, 2015	to	December 8, 2016	181	November 5, 2015	Policy rate raised to 15.5% from 12.5% and interest rate caps removed
	4	December 12, 2016	to	October 31, 2017	179	December 12, 2016	Inflation declined to single digit and expected to remain relatively low, signifying decline in risk premium
	5	November 3, 2017	to	March 15, 2019	250	November 3, 2017	Expectations of further monetary policy loosening following a policy rate reduction in February 2017
Total Obsservations	5	-	-	-	1,192		
	1	April 10, 2012	to	March 18, 2014	234	NA	
Small to Medium	2	March 19, 2014	to	November 1, 2015	256	March 19, 2014	OLF tightened by 600 bps and acess to OLF window restricted to once a week
sized Inter-Bank Market	3	November 3, 2015	to	August 27, 2016	47	November 3, 2014	Policy rate raised to 15.5% from 12.5% and interest rate caps removed
	4	August 31, 2016	to	March 21, 2017	90	August 31, 2016	
	5	March 30, 2017	to	February 11, 2019	128	March 30, 2017	
Total Obsservations	5				755		
	1	June 1, 2012	to	March 19, 2014	72	NA	
Small to Lange sized	2	March 25, 2014	to	August 11, 2014	60	March 25, 2014	OLF tightened by 600 bps and acess to OLF window restricted to once a week
Inter-Bank Market	3	August 18, 2014	to	October 13, 2015	54	August 18, 2014	
Inter-Bank Market	4	October 20, 2015	to	March 17, 2017	58	October 20, 2015	Tightened foreign exchange trading rules
	5	April 20, 2017	to	February 27, 2019	133	April 20, 2017	
Total Obsservations	5	-		-	377		

Table 6a: Structural Breaks relating to Monetary Policy Events

	Event	Period Covered			Observations	Structural Break Dates	Possible Factors		
	1	April 2, 2012	to	March 19, 2014	289	NA			
Medium to Small sized Inter-Bank	2	March 20, 2014	to	November 6, 2015	351	March 20, 2014	OLF tightened by 600 bps and acess to OLF window restricted to once a week		
	3	November 9, 2015	to	October 3, 2016	204	November 9, 2015	Policy rate raised to 15.5% from 12.5% and interest rate caps removed		
warket	4	October 4, 2016	to	August 28, 2017	206	October 4, 2016			
	5	August 29, 2017	to	March 22, 2019	334	August 29, 2017	Policy rate reduced to 11% from 12.5%		
Total Obsservations				-	1,384				
	1	April 2, 2012	to	March 19, 2014	169	NA			
	2	March 20, 2014	to	November 10, 2015	362	March 20, 2014	OLF tightened by 600 bps and acess to OLF window restricted to once a week		
Medium to Medium sized Inter-Bank	3	November 16, 2015	to	May 12, 2016	102	November 16, 2015	Policy rate raised to 15.5% from 12.5% and interest rate caps removed		
Market	4	May 17, 2016	to	August 10, 2017	206	May 17, 2016	Some FX trading rules relaxed		
	5	August 11, 2017	to	March 15, 2019	181	August 11, 2017	Policy rate reduced to 11% from 12.5%		
Total Obsservations	1				1,020				
	1	April 10, 2012	to	March 19, 2014	149	NA			
	2	March 20, 2014	to	August 11, 2014	240	March 20, 2014	OLF tightened by 600 bps and acess to OLF window restricted to once a week		
Medium to Large sized Inter-Bank	3	November 13, 2015	to	May 10, 2016	112	November 13, 2015	Policy rate raised to 15.5% from 12.5% and interest rate caps removed		
Market	4	May 11, 2016	to	March 17, 2017	92	May 11, 2016	Some FX trading rules relaxed		
	5	September 7, 2017	to	February 27, 2019	380	September 7, 2017	Policy rate reduced to 11% from 12.5%		
Total Obsservations					973				

Table 6b: Structural Breaks and Monetary Policy Events: Medium sized banks pricing

	Event	Period Covered			Observations	Structural Break Dates	Possible Factors
	1	April 2, 2012	to	March 19, 2014	251	NA	
	2	March 25, 2014	to	November 3, 2015	286	March 25, 2014	OLF tightened by 600 bps and acess to OLF window restricted to once a week
Large to Small sized Inter-Bank	3	November 4, 2015	to	May 11, 2016	73	November 4, 2015	Policy rate raised to 15.5% from 12.5% and interest rate caps removed
Market	4	May 20, 2016	to	February 23, 2017	103	May 20, 2016	Some FX trading rules relaxed
	5	June 20, 2017	to	March 22, 2019	101	June 20, 2017	Policy rate reduced to 11% from 12.5%
Total Obsservations					814		
	1	April 2, 2012	to	March 12, 2014	261	NA	
	2	March 20, 2014	to	November 3, 2015	272	March 20, 2014	OLF tightened by 600 bps and acess to OLF window restricted to once a week
Large to Medium sized Inter-Bank	3	November 4, 2015	to	May 10, 2016	93	November 4, 2015	Policy rate raised to 15.5% from 12.5% and interest rate caps removed
Market	4	May 17, 2016	to	May 18, 2017	161	May 17, 2016	Some FX trading rules relaxed
	5	June 14, 2017	to	March 22, 2019	219	June 14, 2017	Policy rate reduced to 11% from 12.5%
Total Obsservations					1,006		
	1	April 10, 2012	to	March 18, 2014	156	NA	
	2	March 20, 2014	to	November 11, 2015	248	March 20, 2014	OLF tightened by 600 bps and acess to OLF window restricted to once a week
Large to Large sized	3	November 12, 2015	to	May 12, 2016	74	November 12, 2015 Policy rate raised to 15.5% from 12.5% and in caps removed	
inter-Bank Market	4	May 13, 2016	to	May 15, 2017	107	May 13, 2016	Some FX trading rules relaxed
	5	May 31, 2017	to	March 20, 2019	264	May 31, 2017	Policy rate reduced to 11% from 12.5%
Total Obsservations					849		

Table 6c: Structural Breaks and Monetary Policy Events: Large sized banks pricing

4 Data Sources

The study employed daily data for the period January 2012-March 2019 for which a reliable data set was available. All the data were sourced from the Bank of Zambia covering 14 commercial banks.

Lending data was utilised and include transacting bank, counterparty, loan amount, tenor, interest rate, collateral type, collateral face value, and currency in which the loan was denominated, which were all in the local currency, Kwacha.

Lending transactions are split into intra and cross category trades. The transactions analysed include small to small, small to medium, small to large, medium to small, medium to medium, medium to large, large to small, large to medium and large to large interbank lending. For pricing behavior, lending rates data are used and cover nine trading pairs involving small, medium, and large banks as follows: small-small, small-medium, small-large, medium-small, medium-medium, medium-large, large-large, large-medium, and large-small.

5 Preliminary Findings

According to the exposure matrix results (Table 7), the interbank market in Zambia is incomplete, but not disjointed. An assessment of annual exposure matrices of interbank trades over the sample period (2012-2019) also confirmed the incomplete structure of the interbank money market (Tables a - h in the Appendix).



 Table 7: Interbank Market Exposure Matrix (2009m1-2019m3)

Source: Author's computation using Bank of Zambia database

 $\sqrt{}$ means existence of an active interbank credit line while 0 implies no trade between the corresponding banks

Note: sl- small local banks, ml-medium local sized banks and bl-Big local banks

Further, despite the interbank market being well connected, banks are not completely connected to each other, evidenced by the clustering coefficient of close to 1 (Table 8). Cluster coefficients close to 1 imply that most counterparties of a bank also have credit lines among themselves. This

means that, on average, most counterparties of a bank have trading relationships among themselves (but not all of them). For instance, a bank may have credit lines with numerous counterparties, but this does not necessarily mean that all of these counterparties also have credit lines among themselves. In addition, banks maintained trades with the usual frequent counterparties over the sample period based on the closeness centrality measure reported in Table 7. On average, each bank has acted as a mid-agent for about 1 to 3 banks based on the betweenness centrality measure. This helps with the flow of liquidity in a segmented market.

	Betweenness		Clustering
	Centrality	Closeness Centrality	Coefficient
Overall Sample	1.20000	0.06224	0.92787
2019	3.53333	0.04873	0.70106
2018	2.26667	0.05515	0.84795
2017	2.86667	0.05212	0.73967
2016	2.53333	0.05407	0.77809
2015	1.07143	0.06713	0.89340
2014	1.21429	0.06588	0.87848

 Table 8 Measures of Network Centrality (Annual Average)

Source: Author computations

A further analysis of the interaction of banks reveals that large banks are able to trade (interact) among themselves fully by utilizing all the available credit lines (Table 9). However, their interaction with small and medium banks is limited as they are able to utilize 75% and 83% of the available credit lines on the borrowing side and 70% and 71% on the lending side, respectively. While the interaction among the small-small banks is relatively high, it is, however, limited among the medium-medium pair (Table 9). This implies that large and small banks interact amongst themselves more than the medium sized banks do with the peers.

 Table 9 Utilisation of Credit Lines (%)

		Lend	lers	
		Small	Medium	Large
Borrowers	Small	95	60	70
	Medium	67	50	71
	Large	75	83	100

Source: Author's computation using Bank of Zambia database

In terms of trading preference, large banks tend to borrow from and lend to medium sized banks and among themselves than they do with the small sized banks (Chart 1). This could be attributed to the strict credit risk compliance rules that large and medium sized banks follow when dealing with small banks. On the other hand, large and medium sized banks have more credit lines among themselves and few credit line with small banks. Medium sized banks prefer to trade with large banks than they do with peers and small banks. The preference by small banks is to borrow from medium sized banks than large banks and the peers. Conversely, their lending preference is to the peers (Chart 1).



Chart 1 Lending and Borrowing Preference Indices



Ownership partially explains intense intra-trading within the small and large bank categories. The small bank category is dominated by local banks while the medium and large bank categories are dominated by foreign ownership. The only locally owned banks in the medium and large categories are those in which the Government of Zambia holds shares. Most locally owned banks tend to have more trading linkages among peers. Correspondingly, foreign banks also prefer trading with fellow foreign counterparties. This is consistent with the utilisation of credit lines results (Table 7). Large banks utilize all available credit lines as banks in this category are predominantly foreign owned. Equally, small banks utilize 95% of the credit lines as most of them are locally owned. However, intra category trading among medium banks is the lowest. This is mainly due to the diversity of foreign banks in this category. Most of the foreign banks are Pan-African while others are global multinationals. Generally, global multinational banks have limited and in some cases no credit lines with most local banks unless Government has a stake, and most Pan-African owned banks as these banks are consider as risky counterparties (default risk is higher than most big banks) Therefore, global multinationals prefer to trade mostly with banks in the large category.

In terms of pricing behavior, all the trading pairs among various categories of banks transacted above the policy rate (Table 10). The deviation was even higher in periods when monetary policy was tightened. This was partly to ensure that the pricing of the interbank facility was positive in real terms as the policy rate tended to be lower than the inflation rate at the time. This was coupled with tight liquidity conditions obtaining especially in late 2015 and in 2016 when monetary policy was tightened to contain inflationary pressures. The deviation is more pronounced for the small-large and medium-medium bank pairs where the lending rate exceeded the policy rate by 200 basis points. This could be attributed to limited credit lines between small and large banks as lending by small banks to large banks is usually at a higher premium. This result is in contrast with the evidence by Kim (2017) where large-large banks trade around the target rate and only small-large is below the policy rate while the large-small was above the target rate. On the other hand, the least deviations from the policy rate were among the large-small followed by large-medium pairs. This result is consistent with Muhanga, Mutoti and Zgambo (2009) who established a tendency

for large bank lenders to demand a premium from fellow large bank borrowers compared to small bank borrowers

It is noted that for small-large and medium-medium pairs, the volumes traded tend to be lower than the rest of the pairs. The relatively higher premium over the policy rate for the small to large lending pair than the large to small could be explained in the context of the liquidity cost. This is despite small banks having more credit lines and trading preference with large and medium banks. Large banks tend to generally charge the lowest premium compared to the other categories of banks. This could be explained by their large presence in the retail market where they are able to mobilise deposits at relatively lower cost in addition to having a large network of relationship banking with the corporates. The latter may give them an advantage in the wholesale market outside the interbank that ultimately lowers their liquidity cost.

The Hurst parameter estimates suggest a long-range dependence in interbank lending rates and thus persistence in loan pricing overall: increases in interbank lending rates take a relatively long period followed by an equally relatively long period of decline. This means that the interbank rate is predictable across relationship pairs as the estimated Hurst parameters ranged from 0.640 to 0.818 i.e. there is uniformity in pricing behavior in the interbank money market. This result suggests the absence of price segmentation in the interbank money market contrary to the findings by Muhanga, Mutoti and Zgambo (2009). However, monetary policy actions identified above tend to alter pricing behavior in the interbank market as pricing was characterised by anti-persistence (0 < H < 0.5) and randomness (unpredictable, H = 0.5). For example, when monetary policy was tightened and interest caps⁹ removed in November 2015, the pricing behavior changed and lending rates became unpredictable for the small-medium, small-large, and medium-large pairs given the estimated H parameter value of close to 0.5. While there exists intra-group price segmentation for the large-large pair and mostly for the medium-medium pair across different regimes, it is not the case with the small-small pair. This is due to the Hurst parameter values falling within the same range, suggesting a persistence in pricing, and may interpreted as a case of "colleagueship."

⁹ In 2012 the Bank of Zambia introduced interest caps on lending rates charged by banks and non-bank financial institutions but was, however, discontinued in November 2015.

Table 10a: Pricing Differential

Inter Dents Comment	Pricing Differe	ential (Overall Sample)			Pricing Differe	ntial (Sub-Samples)
Inter-Bank Sagement	mean	Standard Deviation	Structural Break Dates	Regime	mean	Standard Deviation
				1	-0.30	0.94
Small to Small sized			March 21, 2014	2	2.62	3.47
Intor-Rapk Market	1.62	3.42	November 5, 2015	3	5.83	4.77
Inter-Bank Market			December 12, 2016	4	0.35	0.76
			November 3, 2017	5	-0.01	0.31
				1	-0.45	0.92
Small to Medium			March 19. 2014	2	5.06	4.30
sized Inter-Bank	1.68	3.60	November 5, 2014	3	3.72	3.47
Market			May 30, 2016	4	1.02	0.94
			September 8, 2017	5	-0.08	0.36
				1	-0.30	0.75
Small to Large sized	2.24	4.95	March 25, 2014	2	7.15	2.32
Inter-Bank Market	2.21	4.25	August 18, 2014	3	0.25	2.32
			October 20, 2015	4	7.39	4.48
			April 20, 2017	5	-0.09	0.46
				1	-0.36	1.00
Medium to Small			March 20, 2014	2	2.62	3.46
sized Inter-Bank	1.69	3.51	November 9, 2015	3	6.95	4.46
Market			October 4, 2016	4	0.59	0.67
			August 29, 2017	5	0.16	0.31
				1	-0.23	0.97
Medium to Medium			March 20. 2014	2	2.70	3.59
sized Inter-Bank	2.09	3.81	November 16, 2015	3	10.94	1.09
Market			May 17. 2016	4	1.30	1.00
			August 11, 2017	5	-0.06	0.48
				-	0.47	0.07
Madium to Large			Marsh 20, 2014	1	-0.47	0.97
Nedium to Large	1.07	2.07	March 20, 2014	2	3.04	3.90
Sized Inter-Bank	1.97	3.87	November 4, 2015	3	9.93	2.58
Ivlarket			May 11, 2016	4	1.17	1.36
			September 7, 2017	5	-0.10	0.38
				1	-0.63	1.01
Large to Small sized			March 25, 2014	2	2.75	3.76
Inter-Bank Market	1.37	3.42	November 4, 2015	3	3.46	4.29
			May 20, 2016	4	0.90	1.88
			June 20, 2017	5	-0.02	0.47
				1	-0.65	0.94
Large to Medium			March 20. 2014	2	1.78	2.48
sized Inter-Bank	1.41	3.41	November 4, 2015	3	10.35	2.11
Market			May 17. 2016	4	1.08	1.07
			June 14, 2017	5	-0.14	0.56
				-	0.55	2.25
			Marsh 20, 2014	1	-0.55	0.86
Large to Large sized	1 50	2 4 2	Nevember 12, 2014	2	2.20	2.89
Inter-Bank Market	1.39	5.45	May 12, 2016	3	1.49	1.50
			Nav 21 2017	- 4	1.46	0.92
			IVIAY 31, 2017	5	-0.14	0.53

Table 10b: Long Range Dependence Results

Inter Bank Cogement	Hurst Parameter		Structural Break Dates	Beginne	Hurst Parameter
inter-bank Sagement	(Overall Sample)		Structural Break Dates	Regime	(Sub-Samples)
				1	0.565
Small to Small sized			March 21, 2014	2	0.802
Inter Bank Market	0.707	Persistent Process	November 5, 2015	3	0.751
Inter-Bank Market		Fersistent Frocess	December 12, 2016	4	0.780
			November 3, 2017	5	0.774
				1	0.412
Small to Medium			March 19, 2014	2	0.699
sized Inter-Bank	0.818	Persistent Process	November 5, 2014	3	0.468
Market			May 30, 2016	4	0.853
			September 8, 2017	5	0.773
				1	0.463
Small to Large sized			March 25, 2014	2	0.934
Inter-Bank Market	0.679	Persistent Process	August 18, 2014	3	0.408
			October 20, 2015	4	0.945
			April 20, 2017	5	0.673
				1	0.487
Medium to Small			March 20, 2014	2	0.736
sized Inter-Bank	0.768	Persistent Process	November 9, 2015	3	0.911
Market			October 4, 2016	4	0.781
			August 29, 2017	5	0.565
				1	0.442
Medium to Medium			March 20, 2014	2	0.812
sized Inter-Bank	0.77	Persistent Process	November 16, 2015	3	0.776
Market			May 17, 2016	4	0.561
			August 11, 2017	5	0.384
				1	0.537
Medium to Large	0.64	Barristan Barrist	March 20, 2014	2	0.838
sized Inter-Bank	0.64	Persistent Process	November 4, 2015	3	0.522
Market			May 11, 2016	4	0.216
			September 7, 2017	5	0.632
				1	0.562
Large to Small sized			March 25, 2014	2	0.810
Inter-Bank Market	0.646	Persistent Process	November 4, 2015	3	0.769
			May 20, 2016	4	0.076
			June 20, 2017	5	0.989
				1	0.568
Large to Medium			March 20, 2014	2	0.806
sized Inter-Bank	0.646	Persistent Process	November 4, 2015	3	0.702
Market			May 17, 2016	4	0.488
			June 14, 2017	5	0.679
					0.004
			March 20, 2014	2	0.004
Large to Large sized	0.691	Persistent Process	Nevember 12, 2015	2	0.521
Inter-Bank Market	0.051	Feisistent Flotess	May 12, 2016	3	0.350
			May 21 2017	- 4	0.734
			Iviay 51, 2017	5	0.589

6 Conclusion

The segmentation of the interbank money market in Zambia was analysed using network framework analysis, lending and borrowing (trading) preference indices, as well as price differential analysis. The daily data spanned from January 2012 to March 2019.

The study has revealed the existence of segmentation in the interbank market in Zambia. The market structure is incomplete (not all banks have credit lines) but not disjointed as both within and cross bank categories trades occur. Specifically, large banks fully trade with each other by utilizing all the available credit lines. Nonetheless, they charge a relatively higher premium compared to small and medium sized bank borrowers. This result is consistent with Muhanga, Mutoti and Zgambo (2009) where large bank lenders were found to demand a premium from fellow large bank borrowers compared to small bank borrowers. This notwithstanding, large banks interaction with small and medium banks has been limited. The interaction among the small banks with each other is relatively high but commands a relatively lower premium than the case is with lending to medium sized and large banks, indicative of pricing segmentation. However, there is limited interaction among the medium-medium pair and tend to charge each other a relatively higher premium compared to small and large banks. The study also shows that large and small banks interact amongst themselves more than the medium sized banks do with peers. This perhaps could be the reason for high premium charged when medium-medium banks trade among themselves. Further, the paper reveals that trades are also based on ownership. Bank categories with similar or related ownership tend to have more interactions as opposed to bank categories with diverse ownership

Although the results on long-range dependence in pricing show lack of segmentation and predictability in interbank lending rates, the pricing behavior is susceptible to monetary policy stance change which makes lending rates unpredictable for some trading relationship pairs.

It will be interesting to extend this study by considering the role of volume of funds traded and how they may influence pricing behavior of banks in light of the price segmentation established. This is because liquidity in the Zambian interbank market is skewed among the large banks and two medium sized banks.

References

Allen, F. and Gale, D. (1990) 'Incomplete markets and incentives to set up an options exchange', *The Geneva Papers on Risk and Insurance Theory*, 15(1), pp. 17–46. doi: 10.1007/BF01498458.

Allen, F. and Gale, D. (2000) 'Financial Contagion', *Journal of Political Economy*, 108(1), pp. 1–33.

Allen, L. and Saunders, A. (1992) 'Bank window dressing: Theory and evidence', *Journal of Banking and Finance*, 16(3), pp. 585–623. doi: 10.1016/0378-4266(92)90046-3.

Arukaevu, R. (1998) 'The Estonian Money Market', *Russian & East European Finance and Trade*, 34(5), pp. 71–96.

Bech, M. L. and Atalay, E. (2010) 'The topology of the federal funds market', *Physica A: Statistical Mechanics and its Applications*. Elsevier B.V., 389(22), pp. 5223–5246. doi:

10.1016/j.physa.2010.05.058.

Bräuning, F. and Fecht, F. (2017) 'Relationship lending in the interbank market and the price of liquidity', *Review of Finance*, 21(1), pp. 33–75. doi: 10.1093/rof/rfw042.

Cajueiro, D. O. and Tabak, B. M. (2007) 'Time-varying long-range dependence in US interest rates', *Chaos, Solitons and Fractals*, 34(2), pp. 360–367. doi: 10.1016/j.chaos.2006.04.012.

Cajueiro, D. O. and Tabak, B. M. (2009) 'Testing for long-range dependence in the Brazilian term structure of interest rates', *Chaos, Solitons and Fractals*. Elsevier Ltd, 40(4), pp. 1559–1573. doi: 10.1016/j.chaos.2007.09.054.

Chamoli, A., Ram Bansal, A. and Dimri, V. P. (2007) 'Wavelet and rescaled range approach for the Hurst coefficient for short and long time series', *Computers and Geosciences*, 33(1), pp. 83–93. doi: 10.1016/j.cageo.2006.05.008.

Chen, Y., Zhang, L. and Li, K. (2016) 'The Review of Network Theory Applied to the Financial System', *Fifteenth Wuhan International Conference on E-Business*, pp. 1–8.

Chiu, J., Eisenschmidt, J. and Monnet, C. (2019) 'Relationships in the interbank market', *Review* of *Economic Dynamics*. doi: 10.1016/j.red.2019.06.003.

Cocco, J. F., Gomes, F. J. and Martins, N. C. (2009) 'Lending relationships in the interbank market', *Journal of Financial Intermediation*. Elsevier Inc., 18(1), pp. 24–48. doi: 10.1016/j.jfi.2008.06.003.

Colliard, J.-E. *et al.* (2016) 'Overnight interbank markets and the determination of the interbank rate: A selective survey', *International Review of Financial Analysis*. Elsevier Inc., 44(14), pp. 149–161. doi: 10.1016/j.irfa.2016.01.014.

Colliard, J.-E., Foucault, T. and Hoffmann, P. (2016) 'Trading Frictions in the Interbank Market, and the Central Bank'. Available at: http://www.eief.it/files/2016/06/colliard.pdf.

COMESA Monetary Institute (2019) *Guidelines for Analysisng Banking System Interconnectedness and Systemic Risk: Basic Network Analysis.*

Craig, B. R., Fecht, F. and Tümer-Alkan, G. (2015) 'The role of interbank relationships and liquidity needs', *Journal of Banking and Finance*. Elsevier B.V., 53, pp. 99–111. doi: 10.1016/j.jbankfin.2014.12.022.

Gabrieli, S. (2012) 'The Microstructure of the Money Market Before and after the Financial Crisis: A Network Perspective', *SSRN Electronic Journal*. doi: 10.2139/ssrn.1743512.

Gereben, A. (1999) 'The Interbank Money Market in Hungary The Interbank Money Market in Hungary', *National Bank of Hungary, Occasional Papers*, 7(September 1998).

Green, C. *et al.* (2016) 'Overnight interbank markets and the determination of the interbank rate: A selective survey', *International Review of Financial Analysis*. Elsevier B.V., 44, pp. 149–161. doi: 10.1016/j.irfa.2016.01.014.

Ho, T. S. Y. and Saunders, A. (1985) 'A Micro Model of the Federal Funds Market', *The Journal of Finance*, 40(3), pp. 977–988.

Kantelhardt, J. W. (2009) 'Fractal and Multifractal Time Series', *Encyclopedia of Complexity and Systems Sceince*, pp. 3754–3779. doi: 10.1007/978-1-4614-1806-1_30.

Kim, K. (2017) 'A Price-Differentiation Model of the Interbank Market and Its Application to a Financial Crisis', *Finance and Economics Discussion Series*, 2017(065). doi: 10.17016/feds.2017.065.

Kirichenko, L., Radivilova, T. and Deineko, Z. (2011) 'Comparative Analysis for Estimating of the Hurst Exponet for Stationary and Nonstationary Time Series Ludmila Kirichenko, Tamara Radivilova, Zhanna Deineko', *Journal, International ' 'Information Technologies & Knowledge*, 5, pp. 371–388.

Lungu, A. B., Bwalya, A. and Chanda, C. S. (2018) 'An assessment of market segmentation in the interbank market in Zambia', *Bank of Zambia Working Paper*, (01).

Mayordomo, S. *et al.* (2015) 'Fragmentation in the European interbank market: Measures, determinants, and policy solutions', *Journal of Financial Stability*. Elsevier B.V., 16, pp. 1–12. doi: 10.1016/j.jfs.2014.11.001.

Muhanga, I., Mutoti, N. and Zgambo, P. (2009) 'Interbank Money Market Segmentation in Zambia'. Available at: (unpublished).

Oduor, J. *et al.* (2014) 'Segmentation and efficiency of the interbank market and their implication for the conduct of monetary policy', *Working Paper Series, African Development Bank, Tunis, Tunisia*, (202).

Osoro, J. and Muriithi, D. (2017) 'The Interbank Market in Kenya: An Event-Based Stress Analysis Based on Treasury Bill Market', *European Scientific Journal, ESJ*, 13(16), p. 127. doi: 10.19044/esj.2017.v13n16p127.

Rea, W. *et al.* (2013) 'Not all estimators are born equal: The empirical properties of some estimators of long memory', *Mathematics and Computers in Simulation*, 93, pp. 29–42. doi: 10.1016/j.matcom.2012.08.005.

Sichei, M. M., Kiplang, S. and Shimba, C. (2012) 'Segmentation and Efficiency of the Interbank Market in Kenya', *Kenya Bankers Association, Centre for Research on Financial Markets and Policy, Working Papers Series*. (Working Papers Series), (01).

Tabak, B. M. and Cajueiro, D. O. (2005) 'The long-range dependence behavior of the term structure of interest rates in Japan', *Physica A: Statistical Mechanics and its Applications*, 350(2–4), pp. 418–426. doi: 10.1016/j.physa.2004.11.048.

Temizsoy, A., Iori, G. and Montes-Rojas, G. (2015) 'The role of bank relationships in the interbank market', *Journal of Economic Dynamics and Control*. Elsevier, 59, pp. 118–141. doi: 10.1016/j.jedc.2015.07.008.

Zeileis, A. *et al.* (2002) 'strucchange: An R Package for Testing for Structural Change', *Journal of Statistical Software*, 7(2), pp. 1–38. doi: 10.18637/jss.v007.i02.

Appendix



Figure 1 Overnight Interbank Traded Volumes (K billion): 2012-2019





Figure 3 Access to Bank of Zambia Liquidity (OLF Volumes: 2012-2019)



Figure 4 Settlement Balances (Excess Reserves) held at the Bank of Zambia and the Overnight Interbank Rate: 2012-2019



								Lending									
	2012			Small	banks					Mediu	m banks				Large	banks	
		Bank	Bank SL1	Bank SL2	Bank SL3	Bank SL4	Bank SF5	Bank MF1	Bank MF2	Bank MF3	Bank MF4	Bank MF5	Bank ML6	Bank BF1	Bank BF2	Bank BF3	Bank BL4
		Bank SL1		\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	0	0	\checkmark	√	\checkmark	0	\checkmark	0	\checkmark
		Bank SL2	\checkmark		\checkmark	\checkmark	0	\checkmark	0	0	\checkmark	\checkmark	\checkmark	0	0	\checkmark	\checkmark
	Small banks	Bank SL3	\checkmark	\checkmark		\checkmark	0	\checkmark	0	0	0	0	\checkmark	\checkmark	\checkmark	0	\checkmark
		Bank SL4	√	\checkmark		_		\checkmark	0	0	\checkmark	\checkmark	\checkmark	0	\checkmark	\checkmark	\checkmark
	Ba Ba Ba	Bank SF5	0	\checkmark	\checkmark	\checkmark		0	0	0	0	0	0	0	\checkmark	0	\checkmark
		Bank MF1	0	\checkmark					0	0	\checkmark	√	\checkmark	\checkmark	\checkmark	0	√
	B	Bank MF2	0	0	0	0	0	0		0	0	0	0	0	0	0	0
Borrowing	M - diam h h -	Bank MF3	0	\checkmark		\checkmark	0	0	0		0	0	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
	Medium banks	Bank MF4	\checkmark	\checkmark	\checkmark	\checkmark	0	\checkmark	0	0		\checkmark	\checkmark	0	\checkmark	0	\checkmark
		Bank MF5	0	\checkmark		\checkmark	0	0	0	0	\checkmark		\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
		Bank ML6	0	0	\checkmark	\checkmark	0	\checkmark	0	\checkmark	\checkmark	\checkmark		\checkmark	\checkmark	\checkmark	\checkmark
		Bank BF1	0	\checkmark	\checkmark	\checkmark	0	\checkmark	\checkmark	\checkmark	0	\checkmark	\checkmark		\checkmark	\checkmark	\checkmark
	I amar hamba	Bank BF2	0	0	0	\checkmark	0	0	0	\checkmark	\checkmark	\checkmark	\checkmark	0		\checkmark	\checkmark
	Large banks	Bank BF3	0	0	0	0	0	0	0	\checkmark	0	0	0	\checkmark	\checkmark		\checkmark
		Bank BL4	\checkmark	\checkmark	0		\checkmark	\checkmark	0	\checkmark							

Table a: Exposure matrix- Zambian Interbank Market (2012)

Source: Author's computation using Bank of Zambia database

 $\sqrt{means existence of an active interbank credit line while 0 implies no trade between the corresponding banks}$ Note: sl- small local banks, ml-medium local sized banks and bl-large Big local banks Table b: Exposure matrix- Zambian Interbank Market (2013)

									Ler	ding							
	2013			Smal	lbanks					Mediu	m banks				Large	banks	
		Bank	Bank SL1	Bank SL2	Bank SL3	Bank SL4	Bank SF5	Bank MF1	Bank MF2	Bank MF3	Bank MF4	Bank MF5	Bank ML6	Bank BF1	Bank BF2	Bank BF3	Bank BL4
		Bank SL1		\checkmark	\checkmark	0	\checkmark	\checkmark	0	0	\checkmark	\checkmark	\checkmark	0	0	0	\checkmark
		Bank SL2	0		\checkmark	0	0	\checkmark	0	0	\checkmark	0	\checkmark	0	\checkmark	\checkmark	0
	Small banks	Bank SL3	0	0		0	\checkmark	\checkmark	0	0	0	0	0	0	\checkmark	0	\checkmark
		Bank SL4	0	\checkmark	\checkmark		\checkmark	\checkmark	0	0	\checkmark	\checkmark	\checkmark	0	\checkmark	\checkmark	\checkmark
		Bank SF5	0	0	\checkmark	0		0	0	0	0	0	\checkmark	0	\checkmark	0	\checkmark
		Bank MF1	0	\checkmark	\checkmark	\checkmark	\checkmark		0	0	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	0	0
		Bank MF2	0	0	0	0	0	0		0	0	0	0	0	0	0	0
Borrowing	Madium hamba	Bank MF3	0	0	0	0	0	\checkmark	0		0	0	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
	medium banks	Bank MF4	0	\checkmark	\checkmark	0	\checkmark	\checkmark	0	0		\checkmark	\checkmark	0	\checkmark	0	\checkmark
		Bank MF5	0	\checkmark	0	0	0	\checkmark	0	0	\checkmark		0	\checkmark	\checkmark	\checkmark	\checkmark
		Bank ML6	0	0	0	0	\checkmark	0	0	0	0	0		0	\checkmark	0	\checkmark
		Bank BF1	0	0	\checkmark	0	0	\checkmark	0	\checkmark	0	\checkmark	\checkmark		√	\checkmark	\checkmark
	Langahanka	Bank BF2	0	0	0	0	\checkmark	\checkmark	0	\checkmark	0	0	\checkmark	\checkmark		\checkmark	\checkmark
	Large Danks	Bank BF3	0	\checkmark	0	0	0	0	0	\checkmark	0	\checkmark	0	\checkmark	\checkmark		\checkmark
		Bank BL4	0	\checkmark	0	0	\checkmark	0	0	\checkmark	\checkmark	√	\checkmark	\checkmark	\checkmark	\checkmark	

Source: Author's computation using Bank of Zambia database

 $\sqrt{}$ means existence of an active interbank credit line while 0 implies no trade between the corresponding banks Note: s- small banks, m-medium sized banks and l-large banks

Table c: Exposure matrix- Zambian Interbank Market (2014)

									Len	ding							
	2014			Small	banks					Mediu	n banks				Large	banks	
	Bank Bank SL1 Bank SL2 Bank SL3 Bank SL4 Bank SF5									Bank MF3	Bank MF4	Bank MF5	Bank ML6	Bank BF1	Bank BF2	Bank BF3	Bank BL4
		Bank SL1		\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	0	0	\checkmark	0	\checkmark	0	0	0	\checkmark
		Bank SL2	\checkmark		0	\checkmark	\checkmark	\checkmark	0	0	\checkmark	0	\checkmark	0	\checkmark	\checkmark	\checkmark
	Small banks	Bank SL3	\checkmark	\checkmark		\checkmark	\checkmark	\checkmark	0	0	\checkmark	0	\checkmark	\checkmark	\checkmark	0	\checkmark
		Bank SL4	0	\checkmark	\checkmark		\checkmark	\checkmark	0	0	\checkmark	\checkmark	\checkmark	0	\checkmark	\checkmark	\checkmark
		Bank SF5	\checkmark	\checkmark		0		\checkmark	0	0	\checkmark	0	0	0	\checkmark	0	\checkmark
	Bi Bi Bi	Bank MF1	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark		0	0	\checkmark						
		Bank MF2	0	0	0	0	0	0		0	0	0	0	0	0	0	0
Borrowing	Madium hawke	Bank MF3	0	\checkmark	0	\checkmark	0	\checkmark	0		0	0	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
	Medium banks	Bank MF4	0	\checkmark		\checkmark	\checkmark	\checkmark	0	0		0	\checkmark	0	\checkmark	0	\checkmark
		Bank MF5	0	\checkmark	0	\checkmark	0	\checkmark	0	0	\checkmark		\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
		Bank ML6	0	\checkmark	0	0	0	\checkmark	0	0	0	0		\checkmark	0	0	\checkmark
		Bank BF1	0	\checkmark	0	\checkmark	0	\checkmark	0	\checkmark	0	\checkmark	\checkmark		\checkmark	\checkmark	\checkmark
		Bank BF2	0	\checkmark	0	\checkmark	\checkmark	\checkmark	0	\checkmark	√	√	\checkmark				\checkmark
	Large Danks	Bank BF3	0	\checkmark	0	\checkmark	0	\checkmark	0		0	\checkmark		\checkmark	\checkmark		\checkmark
		Bank BL4	0	\checkmark	0	\checkmark	\checkmark	\checkmark	0	\checkmark	√	\checkmark	\checkmark	\checkmark	√	\checkmark	

Source: Author's computation using Bank of Zambia database

 $\sqrt{}$ means existence of an active interbank credit line while 0 implies no trade between the corresponding banks Note: sl- small local banks, ml-medium local sized banks and bl-large Big local banks

Table d: Exposure matrix- Zambian Interbank Market (2015)



Source: Author's computation using Bank of Zambia database

 $\sqrt{}$ means existence of an active interbank credit line while 0 implies no trade between the corresponding banks Note: sl- small local banks, ml-medium local sized banks and bl-large Big local banks Table e: Exposure matrix- Zambian Interbank Market (2016)



Source: Author's computation using Bank of Zambia database

 $\sqrt{}$ means existence of an active interbank credit line while 0 implies no trade between the corresponding banks Note: sl- small local banks, ml-medium local sized banks and bl-large Big local banks

									Len	ding							
	2017			Small	banks					Mediu	n banks				Large	banks	
		Bank	Bank SL1	Bank SL2	Bank SL3	Bank SL4	Bank SF5	Bank MF1	Bank MF2	Bank MF3	Bank MF4	Bank MF5	Bank ML6	Bank BF1	Bank BF2	Bank BF3	Bank BL4
		Bank SL1		\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	0	0	\checkmark	\checkmark	\checkmark	0	0	0	0
		Bank SL2	\checkmark		0	\checkmark	\checkmark	\checkmark	0	0	\checkmark	0	\checkmark	0	0	0	\checkmark
	Small banks	Bank SL3	\checkmark	\checkmark		0	\checkmark	0	0	0	\checkmark	0	\checkmark			0	\checkmark
		Bank SL4	0	\checkmark	0		0	\checkmark	0	0	\checkmark	0	\checkmark	0	0	0	\checkmark
		Bank SF5	\checkmark	\checkmark	\checkmark	\checkmark		\checkmark	0	0	\checkmark	0	\checkmark	0	0	0	\checkmark
		Bank MF1	0	\checkmark	\checkmark	\checkmark	\checkmark		0	0	\checkmark	0	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
	B	Bank MF2	0	0	0	0	0	0		0	0	0	0	0	0	0	0
Borrowing	Modium hanks	Bank MF3	0	0	0	0	0	\checkmark	0		0	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
	Meuluin banks	Bank MF4	0	0	\checkmark	\checkmark	0	\checkmark	0	0		0	\checkmark	0	0	0	0
		Bank MF5	0	\checkmark	√	0	0	0	0	0	\checkmark		0	\checkmark	\checkmark	\checkmark	\checkmark
		Bank ML6	√	\checkmark	√	\checkmark	0	0	0	0	\checkmark	0		\checkmark	0	\checkmark	\checkmark
		Bank BF1	0	\checkmark	\checkmark	0	0	\checkmark	\checkmark	\checkmark	0	\checkmark	\checkmark		\checkmark	\checkmark	\checkmark
	Lange hanks	Bank BF2	0	0	0	0	\checkmark	0	0	\checkmark	\checkmark	\checkmark	0	\checkmark		0	\checkmark
	Large Danks	Bank BF3	0	\checkmark	\checkmark	\checkmark	0	\checkmark	0	\checkmark		\checkmark		\checkmark			\checkmark
		Bank BL4	0	\checkmark	\checkmark	0	0	\checkmark	0	\checkmark							

Table f: Exposure matrix- Zambian Interbank Market (2017)

Source: Author's computation using Bank of Zambia database

 $\sqrt{}$ means existence of an active interbank credit line while 0 implies no trade between the corresponding banks Note: sl- small local banks, ml-medium local sized banks and bl-large Big local banks Table g: Exposure matrix- Zambian Interbank Market (2018)

									Len	ding							
	2018			Small	banks					Mediu	m banks				Large	banks	
	Bank Bank SL1 Bank SL2 Bank SL3 Bank SL4 Bank SF5									Bank MF3	Bank MF4	Bank MF5	Bank ML6	Bank BF1	Bank BF2	Bank BF3	Bank BL4
		Bank SL1		\checkmark	\checkmark	\checkmark	0	\checkmark	0	0	\checkmark	0	\checkmark	0	0	0	0
		Bank SL2	0		√	\checkmark	0	0	0	0	\checkmark	0	\checkmark	0	0	0	0
	Small banks	Bank SL3	0	0		0	0	\checkmark	0	0	0	0	0	0	0	0	0
		Bank SL4	0	\checkmark	0		0	\checkmark	0	0	\checkmark	0	\checkmark	0	0	0	0
		Bank SF5	\checkmark	\checkmark	√	\checkmark		\checkmark	0	0	\checkmark	0	\checkmark	0	0	0	\checkmark
		Bank MF1	0	\checkmark	\checkmark	0	0		0	0	\checkmark	0	0	\checkmark	\checkmark	0	\checkmark
		Bank MF2	0	0	0	0	0	0		0	0	0	0	0	0	0	0
Borrowing	Medium banks	Bank MF3	0	0	0	0	0	0	0		0	\checkmark	0	\checkmark	0	\checkmark	\checkmark
	meuluin banks	Bank MF4	0	\checkmark	√	\checkmark	0	\checkmark	0	0		0	\checkmark	0	0	0	\checkmark
		Bank MF5	0	0	0	0	0	0	0	0	0		0	\checkmark	0	0	0
		Bank ML6	0	\checkmark	\checkmark	0	0	0	0	0	\checkmark	0		\checkmark	\checkmark	0	\checkmark
		Bank BF1	\checkmark	0	\checkmark	\checkmark	0	\checkmark	\checkmark	\checkmark	0	\checkmark	\checkmark		0	0	\checkmark
	Langahanka	Bank BF2	0	\checkmark	√	0	0	0	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark		\checkmark	\checkmark
	Large Danks	Bank BF3	0	\checkmark	0	0	0	\checkmark	0	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	0		\checkmark
		Bank BL4	0	0	\checkmark	0	0	\checkmark	0	\checkmark	\checkmark	\checkmark	0	\checkmark	0	0	

Source: Author's computation using Bank of Zambia database

 $\sqrt{}$ means existence of an active interbank credit line while 0 implies no trade between the corresponding banks Note: sl- small local banks, ml-medium local sized banks and bl-large Big local banks Table h: Exposure matrix- Zambian Interbank Market (2019)

									Len	ding							
	2019			Small	banks					Mediu	m banks				Large	banks	
		Bank	Bank SL1	Bank SL2	Bank SL3	Bank SL4	Bank SF5	Bank MF1	Bank MF2	Bank MF3	Bank MF4	Bank MF5	Bank ML6	Bank BF1	Bank BF2	Bank BF3	Bank BL4
		Bank SL1		\checkmark	\checkmark	0	\checkmark	\checkmark	0	0	\checkmark	0	\checkmark	0	0	0	\checkmark
		Bank SL2	0		0	0	\checkmark	0	0	0	\checkmark	0	\checkmark	0	0	0	0
	Small banks	Bank SL3	0	\checkmark	_	0			0	0	\checkmark	0	\checkmark	\checkmark	\checkmark	0	0
		Bank SL4	0	\checkmark	0		0	\checkmark	0	0	\checkmark	0	\checkmark	0	0	0	\checkmark
		Bank SF5	0	\checkmark	\checkmark	0	_	\checkmark	0	0	\checkmark	0	\checkmark	0	0	0	\checkmark
		Bank MF1	0	0	\checkmark	0	0		0	0	\checkmark	0	0	\checkmark	\checkmark	0	0
		Bank MF2	0	0	0	0	0	0		0	0	0	0	0	0	0	0
Borrowing	Madinum Iaam Ia	Bank MF3	0	0	0	0	0	0	0		0	0	0	0	0	0	0
	Medium banks	Bank MF4	0	\checkmark	\checkmark	0			0	0		0	\checkmark	0	0	0	\checkmark
		Bank MF5	0	0	0	0	0	0	0	0	0		0	\checkmark	\checkmark	\checkmark	\checkmark
		Bank ML6	0	\checkmark	0	0	0	0	0	0	0	0		\checkmark	0	0	\checkmark
		Bank BF1	0	0	0	0	0	0	0	\checkmark	0	\checkmark	0		\checkmark	\checkmark	\checkmark
	Lange heating	Bank BF2	0	\checkmark	\checkmark	0	0	\checkmark		\checkmark	\checkmark						
	Large Danks	Bank BF3	0	0	0	0	0	0	0	\checkmark	\checkmark	\checkmark	0	\checkmark	0		0
		Bank BL4	\checkmark	\checkmark	\checkmark	0	0	\checkmark	0	\checkmark							

Source: Author's computation using Bank of Zambia database

 $\sqrt{}$ means existence of an active interbank credit line while 0 implies no trade between the corresponding banks Note: sl- small local banks, ml-medium local sized banks and bl-large Big local banks